Author's response to reviews

Title: Tetrabenazine as Anti-Chorea Therapy in Huntington Disease: An Open-Label Continuation Study

Authors:

Samuel A. Frank (samuel.frank@bmc.org)
Huntington Study Group (shari.kinel@ctcc.rochester.edu)

Version: 4 Date: 1 December 2009

Author's response to reviews: see over
Thank you for accepting this paper. Below, please see our response (in bold) to the revisions requested.

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

Authors' contributions: Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

The authorship is a group one and should ultimately be listed as is, per the policy of the Huntington Study Group. I spoke with David Botha who informed me to change the authorship to Huntington Study Group (first name) TETRA-HD Investigators (last name) with my email as the primary contact. The individuals and their roles are listed at the end of the document under ‘Author Contributions.’

Figure legends - please provide a description of the data for each file in the figure legends section.

The figure legends have been expanded.

References - please cite all references in numerical order in the text. Reference 22 currently does not appear in the correct position.

Done

Minor revisions

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, "The Title of My Prestigious Publication: A Scientific Study." should be changed to "The title of my prestigious publication: a scientific study."

Done

Author groups - please list the full names of the authors from the author groups in the Acknowledgements section.

Footer - please remove.

Done

Underlining - please remove from the manuscript. You may replace it with bold if you wish.

Done

Abstract - please include the trial registration number in the submission system as well as the manuscript.

Done
Testing centre information - please merge these subsections with the Acknowledgements.
Done

Tables - please remove the visible vertical lines from your tables.
Done

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
The figures were cropped and re-submitted as .tif files.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you